Author's response to reviews

Title: Stuttered swallowing: electric stimulation of the right insula interferes with water swallowing. A case report.

Authors:

Peter Sörös (peter.soros@gmail.com)
Faisal Al-Otaibi (faisalruwais@gmail.com)
Savio W H Wong (waihwong@usc.edu)
J Kevin Shoemaker (kshoemak@uwo.ca)
Seyed M Mirsattari (smirsat2@uwo.ca)
Vladimir Hachinski (vladimir.hachinski@lhsc.on.ca)
Ruth E Martin (remartin@uwo.ca)

Version: 4 Date: 2 February 2011

Author's response to reviews:

Dear Sir or Madam,

as requested, I have made the following changes:

1. The authors in the manuscript are now written exactly as they are in the submission system.
2. Both Figures have been cropped.
3. I have double-checked the spelling.

Thank you for working on my paper.

Regards,

Peter Soros